Author's response to reviews

Title: Retroperitoneoscopic single-site renal biopsy surgery: right indications for the right technique

Authors:

Salvatore Micali (salvatore.micali@unimore.it)
Alessio Zordani (alessio_zordani@hotmail.it)
Riccardo Galli (gallidoc@gmail.com)
Eugenio Martorana (eugeniomartorana@libero.it)
Micaela Piccoli (m.piccoli@ausl.mo.it)
Gianni Cappelli (gianni.cappelli@unimore.it)
Giampaolo Bianchi (bianchi.giampaolo@unimo.it)

Version: 4 Date: 4 September 2014

Author's response to reviews:

1. The procedures described in our manuscript were performed as part of standard care and therefore did not require ethical approval but only a standard consent.
2. Consent: I modified the Materials and methods (line 95 to 100) with the consent statement that all patients subscribed before surgery.
3. I modified the table number 1 to prevent the patients identification.
4. I added the section “authors' contributions” (line 264 to 275)
5. I included the Competing interest form separately
6. The paper was revisited by Dr. Paul Casella, Adjunct Assistant Professor, Division of General Medicine, Department of Internal Medicine, University of Iowa Carver College of Medicine, Iowa City, Iowa.